Author's response to reviews

Title: Efficacy of single dose Gentamicin in combination with Metronidazole versus multiple doses for prevention of post caesarean infection

Authors:

Fadhili Lyimo (fadhili1810@yahoo.com)
Anthony N Massinde (amassinde@gmail.com)
Stephen E Mshana (mshana72@yahoo.com)
Benson Kidienza (benkienya@yahoo.com)
Evelyne Konje (ekonje28@yahoo.com)

Version: 4 Date: 14 March 2012

Author's response to reviews: see over
Dear Editor

Title: Efficacy of single dose Gentamicin in combination with Metronidazole versus multiple doses for prevention of post caesarean infection

Reviewer Comments

I have now had a chance to look at these sample size calculations. The authors have used one of several formulas that can be applied to a comparison of Proportions in a superiority trial. However the authors state that they are designing an equivalence trial. If this is truly the case they should be requested to use one of the sample size formulas that have been developed for an equivalence trial along with a reference.

Our Response

We have reviewed the formula and added a reference to the source where it was obtained. The formula has significantly changed our sample size. { See reference number 32}